Author’s response to reviews

Title: Effectiveness of physical therapy interventions for children with cerebral palsy: a systematic review

Authors:

Heidi Anttila (heidi.anttila@stakes.fi)
Ilona Autti-Ramo (ilona.autti-ramo@kela.fi)
Jutta Suoranta (jutta.suoranta@uta.fi)
Marjukka Makela (marjukka.makela@stakes.fi)
Antti Malmivaara (antti.malmivaara@stakes.fi)

Version: 3 Date: 14 April 2008

Author’s response to reviews: see over
Dear Editor,

As requested, the following formatting changes are made to the manuscript:

1) One website address was removed from the text (page 5, methods section) and a new reference number 31 was created. The reference numbers for the rest of the references were changed accordingly using the Endnote software. In the reference all website addresses are corrected to the requested structure (www.website.com).

2) The title "Figures" is re-titled as "Figure Legends"

In addition, two other small changes were made:

1) One affiliation was added to Marjukka Mäkelä
2) Page 3, background section: The first sentence and the quotation are corrected to "Cerebral palsy (CP) describes "a group of permanent disorders of the development of …"

Yours sincerely,

Heidi Anttila
MSc (Health Sci), PT
Finnish Office for Health Technology Assessment (FinOHTA)
National Development and Research Centre for Welfare and Health (Stakes)
P.O. Box 220, FIN-00531 Helsinki, Finland
Phone: +358 9 3967 2687
Telefax: +358 9 3967 2278
Email: heidi.anttila@stakes.fi